Author's response to reviews

Title: Emerging organisational models of primary healthcare and unmet needs for care: insights from a population-based survey in Quebec province.

Authors:

Jean-Frédéric Levesque (Jean-Frederic.Levesque@inspq.qc.ca)
Raynald Pineault (rpineaul@santepub-mtl.qc.ca)
Marjolaine Hamel (mhamel@santepub-mtl.qc.ca)
Danièle Roberge (daniele.roberge@usherbrooke.ca)
Costas Kapetanakis (ckapetan@santepub-mtl.qc.ca)
Brigitte Simard (bsimard@santepub-mtl.qc.ca)
Alexandre Prudhomme (aprudhom@santepub-mtl.qc.ca)

Version: 4 Date: 4 June 2012

Author's response to reviews:

Dear Editor

We are pleased to submit a revised version of our paper, having taken into account all the reviewer's comments. We have made changes to the manuscript accordingly. This version of the paper has addressed the very insightful suggestions of reviewers. More precisely:

- We have added a sentence stating that the questions were derived from the existing literature and that the questionnaire was validated for face and content validity;

- We have deleted the sentences talking about the collapsing of categories so that it would comply with our recent revisions as per the reviewer's suggestions;

- We have added a sentence in the discussion about the potential increase in type 1 error, as per the reviewer's suggestion;

- We have corrected the first sentence of the discussion. This was a remnant of a previous correction.

- The second reviewer did not ask for revisions.

We hope that this revised manuscript will be judged acceptable for publication and remain fully available to respond quickly if any technical issue would remain.

Best regards.